Author's response to reviews

Title: Knowledge of pressure ulcer prevention: A cross-sectional and comparative study among nurses

Authors:

Mirjam A Hulsenboom (mhulsenboom@gmail.com)
Gerrie JJW Bours (g.bours@zw.unimaas.nl)
Ruud JG Halfens (r.halfens@zw.unimaas.nl)

Version: 3 Date: 8 March 2007

Author's response to reviews: see over
Dear Dr. Saltman,

Please find appended the revised manuscript entitled: “Knowledge of pressure ulcer prevention: a cross-sectional and comparative study among nurses” which I am submitting for publication at the BMC Nursing. Authors of the manuscript are Mirjam A. Hulsenboom, Gerrie J.J.W. Bours and Ruud J.G. Halfens.

Based on the e-mail with requested formatting changes, I changed all mentioned subjects, except for ‘list source of funding’. The study is conducted for a master thesis and therefore funding was not necessary.

I hope you will consider the revised manuscript for publication.

Yours sincerely,

Mirjam A Hulsenboom, MSc RN

Maastricht University
Department of Health Care Studies
Section of Nursing Science
P.O. Box 616
6200 MD Maastricht
The Netherlands
Email: m.hulsenboom@zw.unimaas.nl